Author's response to reviews

Title: Exploring health-related quality of life in eating disorders by a cross-sectional study and a comprehensive review

Authors:

Monica Baiano (monibaiano@virgilio.it)
Pierandrea Salvo (pierandrea.salvo@ulss10.veneto.it)
Pierluigi Righetti (pierluigi.righetti@unipd.it)
Lucia Cereser (lucia.cereser@virgilio.it)
Erika Baldissera (erika.baldissera@ulss10.veneto.it)
Ilenia Camponogara (ilenia.camponogara@ulss10.veneto.it)
Matteo Balestrieri (matteo.balestrieri@uniud.it)

Version: 5 Date: 22 January 2014

Author's response to reviews:

As suggested by the Editor, we put the review first followed by our original findings. Other minor revisions of the text have been added, in order to be consequent to this change.